Author’s response to reviews

Title: “Happiness in the air?” The effects of air pollution on adolescent happiness

Authors:

WenHsu Lin (daniellin@ym.edu.tw)
Wen-Chi Pan (wenchipan@ym.edu.tw)
Chin-Chun Yi (chinyi@gate.sinica.edu.tw)

Version: 3 Date: 28 May 2019

Author’s response to reviews:

1. Verbal Consent

-- Many thanks for confirming in the section 'Ethics Approval and Consent to Participate', that the informed consent obtained was verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

Thank you for this remark. We provided further information on this issue.

2. Reorder Declarations

Please reorder the declarations so that they are in the following format:

Ethics approval and consent to participate
Consent for publication
Availability of data and material
Competing interests
Funding
Authors' contributions
Acknowledgements

Thank you for this reminder. Our declaration is based on this order.